BANCA D'ITALIA

Temi di discussione

del Servizio Studi

Can option smiles forecast changes in interest rates? An application to the US, the UK and the Euro Area

by Marcello Pericoli



Number 545 - February 2005

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board:

Giorgio Gobbi, Marcello Bofondi, Michele Caivano, Andrea Lamorgese, Francesco Paternò, Marcello Pericoli, Alessandro Secchi, Fabrizio Venditti, Stefania Zotteri, Raffaela Bisceglia (*Editorial Assistant*).

CAN OPTION SMILES FORECAST CHANGES IN INTEREST RATES? AN APPLICATION TO THE US, THE UK AND THE EURO AREA

by Marcello Pericoli*

Abstract

This paper evaluates the use of risk-neutral probability density functions implied in 3month interest-rate futures options to assess market perceptions regarding future monetary policy moves; options allow the information content implied in simpler derivatives to be extended by providing indicators for asymmetry and extreme values. First, a cubic spline is implemented to evaluate the densities. Second, the methodology is applied to quotes on deposits denominated in US dollars, euros and sterling from January 1999 to May 2004; results show that markets correctly forecast the monetary easing of 2001 in the United States in the course of the second half of 2000, but not in the euro area and the United Kingdom. The evidence for the tightening cycle of 1999 is mixed: markets expected an increase in euro area policy rates at the beginning of 1999; expectations were less clear for the United States' interest-rate increases. In the case of the United Kingdom the increase was not foreseen.

JEL classification: C52, E58, G13, G14, G15

Keywords: risk-neutral density, cubic spline, monetary policy, interest-rate futures options

Contents

1. Introduction	
2. Stylized facts	
3. A review of methodologies	
3.1 The parametric approach	
3.2 The non-parametric approach	
3.3 Non-parametric estimate	
4. Evaluation of the information content	
4.1 Two case studies	
4.2 A time series analysis	
5. Conclusion	
Tables and figures	
Appendix	
References	

^{*} Banca d'Italia, Economic Research Department.

1. Introduction¹

The widespread use and the availability of data of derivatives on short-term interest rates has widened the scope for analyzing the expectations regarding monetary policy decisions prevailing among market participants. This analysis has recently been developed within the research departments of central banks and is extensively used to extract information about market forecasts and market sentiment on future monetary policy changes.² The use of a simple futures contract with delivery date around the time of the monetary policy meeting has already shown a persistent bias in predicting future interest rates; furthermore, this kind of derivative instrument can only give point estimates of the prevailing market rate at delivery. Alternatively, options written on short-term interest-rate futures contract (hereafter futures options or, more simply, options) can be used to evaluate market expectations regarding the underlying and also to extract additional information on their dispersion, on the probability attached to the occurrence of extreme events as well as on the relative probability of the occurrence of a decrease compared with an increase in the underlying — in statistical terms these three features are characterized by the implied standard deviation, the kurtosis and the skewness of the densities. These features allow confidence intervals to be constructed for the underlying instrument at different delivery dates.³

In order to understand why futures options give the opportunity to obtain confidence intervals it may be useful to consider a simple example. A quote on 14 October 2002 for an American-type call option on the futures contract on the 3-month Euribor with expiration date in March 2003 and discount strike price 95.00, gives the buyer the right to buy the underlying

¹ I would like to thank (in alphabetical order) Carl Chiarella, Neil Cooper, Antonio Di Cesare, Fabio Fornari, Paolo Guasoni, Andrea Lamorgese, Nikolaos Panigirtzouglou, Marco Taboga and seminar participants at the Forecasting Financial Markets 2004 conference for comments and help; obviously I alone am responsible for any errors. The methodology used in this paper was developed during the summer of 2000 when the author was visiting the Bank of England (Monetary Instruments and Market Division) and later at the Economic Research Department of Bank of Italy. The opinions expressed in this paper are those of the author and do not necessarily reflect the position of the Bank of Italy. E-mail: marcello.pericoli@bancaditalia.it

² For an extensive survey of the methodology used in central banks see Bank for International Settlements (1999), Deutsche Bundesbank (1996), Clews et al. (2000). For an application of the predictive power of interest rate futures see Owens and Webb (2001).

³ Even if futures options and futures contracts on 3 month interest rate deposits are widely traded at CME, LIFFE and EUREX, the bulk of derivatives on these instruments is still traded in the over-the-counter market where interest caps and interest floors take the lion's share. The main difference between an OTC and a regulated market is that in the former derivatives are traded with a custom-made expiry date, while in the latter the expiry date is fixed.

futures at any time before expiration at 95.00 (e.g. 100.0-95.0=5.0 per cent). Usually, every day such options are quoted for several discount strike prices, e.g. 95.125, 95.25, 95.50 etc., and for different expiration dates. The tenet in option pricing for market practitioners is given by the original Black-Scholes-Samuelson formula, which simply assumes that the underlying asset follows a geometric Brownian motion with constant volatility and, hence, the implied probability density function is log-normal.⁴ However, even if the Black-Scholes-Samuelson formula is widely used, each currently observed option shows different volatility, which tends to be larger as the strike price moves further away from the underlying spot price (this phenomenon is known as the smile of the option). Thus, the market assigns a higher quote to the option if, for a given volatility of the underlying, it considers the occurrence of the corresponding strike at the expiration date to be more likely. This creates the occurrence of positive changes in the underlying compared with negative changes — and of fat-tails in the distributions — since extreme events can be considered more likely.

There is a pervasive practice among practitioners to price options in terms of implied Black-Scholes-Samuelson volatility instead of market prices, i.e. units of volatility corresponding to options prices through the Black-Scholes-Samuelson formula. This practice can be regarded as inconsistent since, on the one hand, markets quote options in terms of Black-Scholes-Samuelson implied volatility and, on the other, volatility is not constant across strike prices. Different theoretical models have been proposed to explain the 'smile effect' but, more interestingly for policy-makers and asset managers, the estimate of the connected probability density function documents the beliefs of market participants about the underlying data generating process (under the non trivial assumption of risk-neutrality). Then, the shape of the smile provides an evaluation of the probability distribution that market participants assign to the occurrence of a strike price at a certain date given the underlying spot price.⁵

⁴ If C and C^{BS} are the observed and the Black-Scholes-Samuelson option prices respectively, the implied volatility γ is defined as the solution to $(C^{BS}(S, X, \tau, \gamma, r) = C)$ where the terms in brackets are the underlying price, the strike price, the time to maturity, the volatility and the risk free rate, respectively.

⁵ Another feature of observed options is that historical volatility is usually lower than the lowest implied volatility (which is usually observed for around at-the-money options). The difference between historical volatility and at-the-money implied volatility is often used as a first tool to assess market sentiment. In this sense, when this difference increases, markets usually show higher kurtosis in their probability density functions and assign a larger probability to the occurrence of extreme events.

When this methodology is applied to derivatives on short-term interest rates it is straightforward to use the 'smile effect' as a tool to assess the predictive power of market participants in forecasting changes in policy interest rates at periodical central bank meetings.

The methodology developed here can be used to gauge whether, in the United States, in the euro area and in the United Kingdom, derivatives markets have accurately predicted monetary policy changes. Furthermore, these tools make it possible to test additional hypotheses as to whether policy-makers have driven expectations or, alternatively, have been driven by market sentiment.⁶

The time series of probability density functions (hereafter pdfs) implied in futures options written onto 3-month interest rates on deposits denominated in US dollars, euros and sterling may therefore provide relevant information about market expectations concerning the moves of the monetary authorities before their periodical meetings. The aim is to assess the degree of market participants' accuracy in anticipating interest rate changes as well as to gauge the impact of different communication strategies which can differently shape investor sentiment. Statistics extracted from these pdfs are used to evaluate their predictive power of directives expected at the Federal Open Market Committee (FOMC) in the United States, at the Governing Council meeting (GC) in the euro area and at the Monetary Policy Committee (MPC) in the United Kingdom, respectively, from January 1999 to May 2002.⁷

The novelty of this paper is the use of pdfs to assess market perceptions of monetary policy moves. In fact, the previous literature has only incidentally analyzed the predictive

⁶ According to so-called central-bank watchers, a comparative analysis of monetary policies in the United States and in the euro area shows that, since 1999, ECB interest-rate changes have not been completely anticipated by markets, while Federal Reserve changes were fully anticipated until the reversal of monetary stance in January 2001. Since 2001, monetary authorities in the United States, in the euro area and in the United Kingdom have substantially eased their monetary stance, driving short-term rates to historical lows. However, according to market commentary, the moves of the Federal Reserve have been different from those of the ECB since, in the US market, agents appear to have anticipated the decisions of the Federal Reserve and seem to have driven interest rates towards the desired values. This empirical regularity may either reflect a deliberate decision of the Federal Reserve to influence expectations or a conditioning power of markets on the moves of monetary authorities. Even if the direction in this causality relationship is uncertain given the endogeneity of the link between policy and short-term rates, it is interesting to compare the US with the euro area where, conversely, this phenomenon has not been observed, thus drawing criticism of the communication strategy adopted by the ECB.

⁷ The FOMC is held eight times a year, although the President of the Federal Reserve system may call for an intra-meeting in order to introduce additional policy-rate changes. The Governing Council of the European Central Bank is held twice a month; since 8 November 2001 decisions on policy rates are taken only at the first meeting of the month. The Monetary Policy Committee of the Bank of England lasts two days and is held every Tuesday-Wednesday or Wednesday-Thursday during the first week of every month.

power of forward-looking securities with respect to interest-rate changes. Söderlind and Svensson (1997) and Fornari and Violi (1998) compare the pdfs of long term government bond yield between different dates to extract also expectations regarding interest-rate changes; Joundeau and Rockinger (2000) compare different methods of evaluating pdfs of exchange rates at two dates corresponding to tranquil and turbulent periods in the foreign exchange market, respectively. The Bank of England (2004, p. 9) presents charts with time series of the skewness of the 6-month implied skew from interest-rate options for the US dollar, the euro and the pound sterling.

I use a cubic spline to estimate the smile, hereafter defined as the relationship between the volatility and the delta of the options. *Pdfs* are then computed with the Rubinstein methodology by differentiating twice the estimated call with respect to the strike prices. Estimates use end-of-day settlement data on three regulated markets: 3-month interest-rate futures options are traded at CME, LIFFE and EUREX.

The paper is organized as follows. Section 2 presents the main stylized facts in the money markets of the United States, the euro area and the United Kingdom. Section 3 reviews the methodologies used to estimate pdfs implied in options and the sub-section 3.3 presents the non-parametric method I use. Section 4 shows the results; first a comparison among pdfs at different dates around selected central bank meetings is presented; second, the time series of some relevant pdf statistics are analyzed. Section 5 concludes. A description of the data is presented in the Appendix.

2. Stylized facts

Since 1999 policy and short-term rates in the three areas have moved in synchrony, with the United Kingdom slightly anticipating the other two areas. Central bank decisions for the three areas are presented in Tables 2-4 in the Appendix. In the United States, from July 1999 the Federal Reserve steadily raised the policy rate (federal funds target rate) by 1.75 percentage points up to 6.5 per cent in less than twelve months; rates were left unchanged during the second half of 2000 when doubts arose about the strength of growth of the US economy; from January 2001 until year end the policy rate was cut eleven times, by 4.75 percentage points, down to 1.75 per cent, which is the lowest level since the 1950s. Table 1 also shows the 'bias'

of the Federal Reserve (or since February 2000 the 'balance of risks'), that is a forward-looking assessment of the likely future monetary moves, released jointly with the decision on interestrate changes; it ranges from 'easy' to 'neutral' and to 'tighter'. In the euro area, the policy rate (the rate on main refinancing operations) was fairly steady in 1999 and frome March 2000 was increased by 1.75 points to 4.75 per cent; from March 2001 the rate was cut four times until year end, by 1.5 points. In the United Kingdom, the policy rate (rate on repurchase agreement operations) was reduced by one point to 5 per cent between January and September 1999; thereafter, the buoyant British business cycle forced the monetary authority to raise the policy rate back to 6 per cent by March 2000, months before the increase implemented in the United States and in the euro area. Between March and December 2001 the rate was cut seven times by a total of 2 percentage points.

Three-month interest rates followed policy rates in the three areas. In the last quarter of 1999, when a quick tightening of monetary stance was in place, the slope of the money market curve — measured by the difference between the 3-month and the policy rate — significantly steepened, signalling expectations of further increases in future policy rates. The slope decreased substantially from January 2000, became flat during 2000 and inverted somewhat at the beginning of the rate cuts. What the money market slope documents then is just a mechanical textbook example: it is steep during phases of tightening and flat or inverted during monetary stance easing.

Similarly, futures contracts on short-term interest-rate deposits in the three areas have not provided good forecasts of future monetary policy moves for any areas.

From 1999 until the end of 2000, market participants assigned to the President of the Federal Reserve a superior information knowledge with respect to that of the European Central Bank Council. According to the financial press and newsletters of the major investment banks, in 1999 and 2000, in the US market the monetary authority drove short-term rates towards the desired values. This phenomenon could be the outcome of the communication strategy adopted by the Federal Reserve and of the transparency perceived by markets through the press communiqués from 1994; on the other hand, this could be the outcome of a leading effect of money markets on the choice of the central bank. However, such a causal relationship is hard to identify owing to endogeneity problems in the link between policy and short-term rates. Conversely, in the euro area, where the judgement of analysts and market participants

regarding monetary policy decisions has been more less favourable, critics have frequently pointed out the failure of the monetary policy communication strategy, which has often tried to surprise the markets with policy-rate changes.

The use of *pdf*'s implied in futures options may give useful insights for assessing this issue and evaluating the sentiment of market participants about central bank moves and their credibility.

3. A review of methodologies

In this paper I use the following standard terminology:

- -S is the underlying spot price;
- -X the strike price;
- $-\phi(S)$ the risk-neutral *pdf*, $\Phi(S)$ the risk-neutral cumulative distribution function (hereafter *cdf*);
- f(S) the normal *pdf*, F(S) the normal cumulative distribution function;
- -t, T and $\tau = (T t)/365$ are, respectively, the current period, the expiration date and the time to expiration as a percentage of the year;
- r is the risk-free interest rate and $e^{-r\tau}$ is the non-stochastic time-homogenous discount factor in continuous time between t and T;
- C the current price of a call option and P the current price of a put option with payoff at time T given by $(S_T - X)^+$ and $(X - S_T)^+$, respectively, where $(y)^+$ stands for $\max(0, y)$;
- C^{BS} the Black-Scholes-Samuelson price of a call option is

$$C^{BS} = S \cdot F\left(\frac{\ln\frac{S}{X} + (r + \sigma^2/2)\tau}{\sigma\sqrt{\tau}}\right) - e^{-r\tau}X \cdot F\left(\frac{\ln\frac{S}{X} + (r - \sigma^2/2)\tau}{\sigma\sqrt{\tau}}\right)$$

- σ is the implied volatility, e.g. $\sigma = \arg \min_{\sigma} |(C C^{BS})|;$
- Δ is the delta of the option, i.e. the sensitivity of the call to the underlying $(\partial C/\partial S)$, which in a Black-Scholes-Samuelson world is given by $F(d_1)$, where $d_1 = \frac{\ln \frac{S}{X} + (r + \sigma^2/2)\tau}{\sigma\sqrt{\tau}}$;
- \mathcal{V} is the vega of the option, i.e. the sensitivity of the call to the volatility $(\partial C/\partial \sigma)$, which in a Black-Scholes-Samuelson world is given by $S \cdot f(d_1)\sqrt{\tau}$;

- I assume that the *put-call parity* holds, i.e. there are no arbitrage opportunities between a portfolio formed by one call option and $Xe^{-r\tau}$ of cash and one portfolio formed by one put option and one unity of its underlying, namely $C + Xe^{-r\tau} = P + S$. Then, put options can be transformed into call options through $C = P + S - Xe^{-r\tau}$.

During the last ten years the literature on financial econometrics and computational methods in finance has extensively used the analysis of *pdf*s implied in asset prices to go beyond the simple Black-Scholes-Samuelson option pricing formula, a tenet in option What the Black-Scholes-Samuelson model assumes is a constant variance for pricing. different degree of moneyness.⁸ Conversely, it is observed that options traded with different degrees of moneyness have different volatilities and, moreover, that the assumption that the underlying follows a simple or a geometric Brownian motion is too naive and does not give a reasonable description of his distribution. In reality the volatility of the options has a nonlinear relationship with the strike: call options show a negative relationship with their implied volatility, while put options show a positive relationship. Thus, what is actually thought of as a smile effect around the underlying spot price is the left tail of the call option schedule and the right tail of the put option schedule, which correspond to out-of-the-money (OTM) calls and puts, respectively. Implied volatility is as high as the strike is far from the underlying, both OTM calls and OTM puts, and the minimum value is usually observed for OTM calls — see Figure 1.

⁸ We define the moneyness as the percentage spread of the strike price, X, from the underlying spot price, S, e.g. (S - X)/X. Usually the moneyness can be measured by the delta (Δ) of the option: in-the-money options have $\Delta \in (0, 0.50)$, at-the-money options have $\Delta = 0.50$ and out-of-the-money options have theoretically $\Delta \in (0.5, \infty)$. In practice, one considers $\Delta = 0.25$ for in-the-money options and $\Delta = 0.75$ for out-of-the-money options, see Malz (1997).



VOLATILITY SMILE

Figure 1

Estimates of *pdf*'s have strong advantages with respect to surveys, market polls and simple futures strip analyses. First, *pdf*'s can be used to give not only point estimates, but also confidence interval estimates for the underlying price. Second, they show how market expectations shift and, through the joint analysis of futures contracts at different expiration dates, provide an evaluation of market sentiment. Last, their higher moments can add further insights about the dispersion of market participants' beliefs.

However, *pdf*'s are estimated under the assumption of risk-neutrality and, therefore, cannot be directly related to 'objective market probabilities', although a precise mathematical relation between them exists. In other words, the probability measure governing the risk-neutral estimates implies that the movements of discounted underlying asset prices are martingales, which is not true under the objective measure. Hence, in this framework the underlying asset price does not depend upon the subjective preferences of market participants. A way to compare the two densities and the risk aversion function is given in Aït-Sahalia and Lo (1998).⁹ Analogously, Coutant (1999) establishes a relationship between subjective and risk-neutral *pdf* using an approximation with Hermite polynomials. What does this imply

⁹ Let $M_{t,T}$ be the stochastic discount factor between t and T defined as the ratio between the marginal utilities at period T and t, Θ and θ the subjective distribution and density functions, respectively, defined over the entire domain of the asset price **S** and E_{Θ} the connected expectation operator. The pricing kernel is then given

for risk-neutral pdf estimates? Approximately, one can say that the true objective pdf will be slightly shifted towards the right with respect to the risk-neutral estimate. Söderlind and Svensson (1997) show that, under the assumption that the asset price and the stochastic discount factor are jointly log-normally distributed, the objective pdf can be obtained by simply shifting to the right the risk-neutral pdf by the amount equal to the covariance between the asset price and the stochastic discount factor. Aït-Sahalia et al. (2001) document how to extract subjective pdfs without introducing any assumption about the risk aversion function.

The literature presents two kinds of approaches in the *pdf* estimates: the parametric and the non parametric approach.

3.1 The parametric approach: stochastic volatility and mixture of log-normals

The introduction of stochastic volatility in the Black-Scholes-Samuelson framework allows the smile effect to be modelled parametrically. The simplest and tractable version of stochastic volatility assumes the existence of a regime switching between two states of the world which can be thought of as high and low volatility states. Then volatility and the drift terms in the geometric Brownian motion are expected to follow a simple Bernoulli process with probability p. I also assume that there is no correlation between the Brownian motion and the probability of being in a high or low volatility state. This framework allows skewed smiles to be built. Under the risk-neutral probability measure, the asset follows the standard

by

$$S_t = E_{\Theta} \left[M_{t,T} \cdot (S_T - X)^+ \right]$$
$$= \int_{S \in \mathbf{S}} M_{t,T} \cdot (S_T - X)^+ d\Theta(S)$$

and the corresponding risk-neutral pdf by

$$\phi(S) = \frac{M_{t,T}}{\int_{S \in \mathbf{S}} M_{t,T} \cdot d\Theta(S)} \theta(S)$$

Then, the risk aversion function and the utility function, which determine the stochastic discount factor, are a key factor in this relationship between subjective and risk neutral. Without introducing ad-hoc assumptions about the form and the distribution of the stochastic discount factor it is not possible to recover a precise link between the two functions.

stochastic differential equation that follows

(1)
$$\frac{dS_t}{S_t} = u_t \cdot dt + \gamma_t \cdot dW_t (u_t, \gamma_t) = \begin{cases} (u_1, \gamma_1) & \text{with } \Pr = p, \ \forall t \\ (u_2, \gamma_2) & \text{with } \Pr = 1 - p, \ \forall t \end{cases}$$

where u_t and γ_t are the risk-neutral drift and the variance of the process, respectively, dW a standard Wiener process. This is basically a Black-Scholes-Samuelson world with two states of nature, and then the Black-Scholes-Samuelson formula applies in each state of nature and the final price of the option is an average of the two prices weighted by the probability of the occurrence of a particular state of nature, i.e. $C^{BS}(u_t, \gamma_t) = p \cdot C^{BS}(u_1, \gamma_1) + (1-p) \cdot C^{BS}(u_2, \gamma_2)$. The parameter vector to be estimated is then $\rho = (u_1, u_2, \gamma_1, \gamma_2, p)$. The mixture of two log-normal densities can be found through an optimization algorithm by minimizing the error between current and Black-Scholes-Samuelson option prices. This procedure aims to minimize the squared differences between the current price and that implied in its mix-density given by the Black's approximation, namely

$$\varepsilon = C - e^{-r\tau} \sum_{i=1}^{2} p_i \left\{ e^{\ln S + \left(u_i - \frac{\gamma_i^2}{2}\tau\right)} \cdot F\left[d_1\left(i\right)\right] - X \cdot F\left[d_2\left(i\right)\right] \right\},$$

where $p_i \in (0, 1)$, $p_1 + p_2 = 1$, $d_1(i) = \left[\ln \frac{S}{X} + \left(u_i + \frac{\gamma_i^2}{2} \tau \right) \right] / \sqrt{\gamma_i^2 \tau}$, $d_2(i) = d_1(i) - \sqrt{\gamma_i^2 \tau}$. Parameters are given by $\rho = \arg \min_{\rho} \sum_{i=1}^{N_O} \varepsilon_i^2$, where N_O is the number of options.

Given the estimated parameter values, option price *pdf*'s can be simulated for different strike prices. Applications of this method are used in Bahra (1996), Fornari and Violi (1998), Melick and Thomas (1997) and Pericoli (2000).

3.2 The non-parametric approach

Under the risk-neutral probability measure with $pdf \phi$ and $cdf \Phi$, assuming constant interest rate and that S follows a Markov process, the price at time t of a call futures option C

with expiration at T and payoff $(S_T - X)^+$ is

(2)

$$C_{t} = e^{-r\tau} \cdot E\left\{(S_{T} - X)^{+} | S = S_{t}\right\}$$

$$= e^{-r\tau} \int_{-\infty}^{\infty} (S - X)^{+} \cdot d\Phi(S)$$

$$= e^{-r\tau} \int_{X}^{\infty} (S - X) \cdot \phi(S) dS$$

analogously the price of a put option can be converted into a call option through the call-put parity and defined in the same way. Then the implied risk-neutral *pdf* is obtained by double differentiating equation (2) with respect to X,¹⁰ namely

(3)
$$\phi(S_T) = e^{r\tau} \left. \frac{\partial^2 C}{\partial X^2} \right|_{X=S_T}$$

Hence, numerical differentiation of the price of a call with respect to the strike price gives an estimate of the implied *pdf*. Methods based on equation (2) are employed by Aït-Sahalia and Lo (1998), Malz (1997) and Neuhaus (1995). Cooper (2000) and Bliss and Panigirtzouglou (2002) show through a Montecarlo simulation the superiority of the non-parametric approach over the parametric one, since the former is more stable and invariant to outliers.

The basic difference between the two approaches is given by the assumption about the distribution of the short-term interest rate given by equation (1). Model (1) is unrealistic for interest rates; more reasonable models which encompass mean reverting properties or the possibility of modelling the term structure and the volatility term structure in a Heath-

$$\frac{\partial}{\partial X} \left(e^{-r\tau} \int_X^\infty (S - X) \cdot \phi(S) dS \right) \Big|_{X = S_T} = e^{-r\tau} \left(\underbrace{\left[(S - X) \cdot \phi(S) \right]}_{=0} + \int_X^\infty \phi(S) dS \right) = e^{-r\tau} \int_X^\infty \phi(S) dS$$

and the second

$$\frac{\partial}{\partial X} \left(e^{-r\tau} \int_X^\infty \phi(S) dS \right) \Big|_{X=S_T} = e^{-r\tau} \left(\phi(S) - \int_X^\infty \underbrace{\frac{\partial \phi(S)}{\partial X}}_{=0} dS \right) = e^{-r\tau} \phi(S)$$

¹⁰ By applying the Leibniz-Newton rule to equation 2, the first derivative is

Jarrow-Morton framework would be more interesting.¹¹ However, when one departs from the simplest models, estimates become very complicated and the larger structure does not allow much flexibility in the *pdf* shape, which sometimes becomes unrealistic. On the other hand, the parametric approach provides estimates of the parameters which are necessary when one is pricing the option through a Feynman-Kac partial differential equation. However, when the *pdf* is uniquely intended as a tool to convey information on market expectations, the non-parametric approach can be more useful and more flexibile. In this work the focus is on the information content of the *pdf* s and I use the latter approach to exploit its greater flexibility.

3.3 Non-parametric estimate

My methodology departs somewhat from the standard non-parametric methods in order to gain stability in its estimates: I move from the $[\sigma, X]$ space to the $[\sigma, \Delta]$ space since in the former the smile is very sensitive to deep ITM options. The rationale behind this choice can be seen in Cooper (2000). Estimates are done through the following steps:

A. Quotes of put options written on interest-rate futures contracts are taken every day for a given time horizon. In order to use the most liquid contracts only in-the-money (ITM) and at-the-money (ATM) option quotes are used since they are more liquid; moreover, only put options are used since, given the discount quotes, they are equivalent to call options, which are more liquid in this type of market, i.e. a put on a futures with discount strike 96.50 is equivalent to a call on an interest-rate futures with strike 3.50 per cent. Note that for futures options the Black-Scholes-Samuelson formula for a call is given by (see Hull 2000)

$$C^{BS} = e^{-r\tau} \left[S \cdot F\left(\frac{\ln(S/X)}{\sigma\sqrt{\tau}} + \frac{\sigma\sqrt{\tau}}{2}\right) - X \cdot F\left(\frac{\ln(S/X)}{\sigma\sqrt{\tau}} - \frac{\sigma\sqrt{\tau}}{2}\right) \right]$$

where S and X are redefined as 100 minus the futures contract and the strike, respectively.

¹¹ The simplest extension of equation (1) is given by the Ho-Lee model which emcompasses the basic features of the Heath-Jarrow-Morton framework. Let f(t,T) be the instantaneous forward interest rate between t and T, then the stochastic differential equation for the short term rate is given by $dr_t = \left(\frac{\partial f(0,T)}{\partial T}\Big|_{T=t} + \gamma^2 t\right) dt + \gamma dW_t$. Note that when one considers only the *pdf* for one interest rate at a given point in the future, t becomes a constant and the only difference from equation (1) is given by the slope of the forward curve between 0 and t. For an extensive application of parametric estimates in the Heath-Jarrow-Morton framework see Amin and Ng (1997).

- B. The implied volatility, σ , for each option is found by minimizing through a Newton Rapson method the module of the distance between the call price given by the Black-Scholes-Samuelson formula and the current call price weighted by its vega, i.e. $\sigma = \arg \min |(C^{BS} - C) / \mathcal{V}|$.¹² American-type options are corrected with the Barone-Adesi-Whaley methodology to apply the Black-Scholes-Samuelson formula. I then obtain the volatility smile, i.e. the volatility as a function of the strike, $\sigma = g(X)$.
- C. I pass from the $[\sigma, X]$ space to the $[\sigma, \Delta]$ space by considering the approximation $\Delta = F(d_1)$. Put options with Δ larger than 0.9 and smaller than 0.1 are not considered since very deep-ITM and deep-OTM futures options are very illiquid and can unnaturally stress the skewness of the *pdf*. The passage from the $[\sigma, X]$ space to the $[\sigma, \Delta]$ space is motivated by its stability with respect to the former.¹³ Figure 2 suggests that an evaluation of the smile through the strike/volatility (or also the strike/moneyness) schedule would overweight the tails of the distributions, signalling the existence of humps for very far from the money quotes. Alternatively, the delta/volatility schedule, cleaned of extreme values, would give more 'reasonable' shapes for the *pdf*.
- D. The schedule between σ and $\Delta = \delta(\sigma, X)$ is built by means of numerical interpolation. The volatility is given by

$$\widehat{\sigma} = \arg\min\left[\eta(\sigma, \delta(\sigma, X)) - \sigma\right]^2$$

where η is a cubic spline which gives a smooth piecewise polynomial interpolation between σ and $\Delta = \delta(\sigma, X)$ weighted by their vega $\mathcal{V}(X, \sigma)$ with 1,000 points.

- E. Estimated volatility is plugged into the Black-Scholes-Samuelson formula to obtain the option price, namely $\widehat{C} = C^{BS}(X, \widehat{\sigma})$.
- F. The central second order derivatives of $\widehat{C}(X)$ give the risk-neutral *pdf*, namely $e^{r\tau} \cdot \partial^2 \widehat{C} / \partial X^2 \Big|_{X=S_T} = \phi(S).^{14}$

¹⁴ The central second order derivative is defined as $\partial^2 C / \partial X^2 = [C_{i+1} - 2C_i + C_{i-1}] / (step)^2$, where *step* is the partition of the exercise price scale. This approximation is of order $O\left[(step)^2\right]$.

¹² Similarly, the implied volatility given by the squared difference between the current and the BS price, $\sigma = \arg \min \left[\left(C^{BS} - C \right) / \mathcal{V} \right]^2$, gives similar results.

¹³ Some authors consider the $[\sigma, X/S]$ space where X/S is the distance from its strike, i.e. its moneyness, instead of the $[\sigma, \Delta]$ space.

Figure 2



SMILE AND DENSITY FUNCTION

deita/volatility







The time series of the statistics of the pdfs can be used to analyze changes in monetary policy stance in the three areas.¹⁵

A further caveat is in order. The Black-Scholes-Samuelson option formula is supposed to hold for European-type options while options on 3-month interest-rate futures are of the American type, i.e. they can be exercised at any time before expiration. A correction must then be carried out to eliminate this bias. In this paper the Barone, Adesi and Whaley (1987) correction is implemented. Alternatively, Fornari and Violi (1998) and Malz (1997) show Black-Scholes-Samuelson estimates of American-type option pdfs where a lower bound is given for European-type pdf estimates.

In order to compare the statistics, estimated pdfs should be re-scaled to a common expiration date. In fact, the shape of the pdf for options traded on 15 May 2002 with expiration date 16 September 2002 — with τ equal to 0.34 — is different from that of options traded on 29 May 2002 with the same expiration date — with τ equal 0.30; more precisely, longer dated options tend to be more dispersed around the mean and have larger kurtosis. The Bank of England (2001) adjusts the estimated pdfs by interpolating the volatilities along the maturity spectrum of the options. In this paper a simpler correction is implemented; the implied volatility is divided by $\sqrt{\tau}$ at point 4 above. It can be shown that this method is empirically equivalent to the one employed by the Bank of England (2001).¹⁶

¹⁵ The statistics commonly used are: *i*) the mean which is the expected value of the distribution and by construction coincides with the futures current price, *ii*) the mode which is the most likely outcome, *iii*) the median which is the value that assigns a 50 per cent probability to the occurrence of that outcome, *iv*) the skewness which characterizes the distribution of probability on either side of the mean (a positively/negatively skewed distribution is one for which there is more/less probability attached to outcomes higher than the mean than to outcomes below the mean; a normal distribution has a skew of zero), *v*) the Pearson skewness which is given by (mean-median)/standard deviation and has a similar interpretation to the simple skewness coefficient, *vi*) the kurtosis which measures the fat tails or alternatively the peak of a distribution (it also measures the likelihood of extreme outcomes, the fatter the tails of the distribution and the more peaked it is around the mean; the normal distribution has a kurtosis equal to three). Note that kurtosis can also be defined by $[0.5 \cdot (\sigma_{0.25} + \sigma_{0.75}) - \sigma_{0.50}] / \sigma_{0.50}$ and skewness by $(\sigma_{0.25} - \sigma_{0.75}) / \sigma_{0.50}$, where σ_i stands for the implied volatility computed for an option with Δ equal to *i*. See Malz (1996 and 1997) for an application to currency options.

¹⁶ In the example above, the volatilities of the first options are divided by $\sqrt{0.34}$, that of the second options by $\sqrt{0.30}$.

Refiniments

Following Aït-Sahalia and Lo (1998) the relationship between Δ and σ has been estimated through a Epanitchev-kernel with optimal bandwidth; call options have also been introduced in addition to put options; estimates for different time horizons have been performed. Results do not differ.

4. Evaluation of the information content

4.1 Two case studies

As a case study I compare the pdfs for the 3-month interbank eurodeposit interest rate denominated in dollars, euros and sterling around the start of an inversion of the monetary policy stance. In particular, the tightening cycle started at the end of June 1999 in the United States, on 4 November 1999 in the euro area and on 8 September 1999 in the United Kingdom; the easing cycle started on 3 January 2001 in the United States, on 10 May 2001 in the euro area and on 8 March 2001 in the United Kingdom. For three dates around these episodes (two before and one after) pdfs are estimated with a constant 90-day horizon – see Figures 3,4,5.

Phases of monetary tightening

In the United States the market was taken completely by surprise by the increase decided at the FOMC on 29 June 1999, which, furthermore, in the following days was perceived as isolated and not as the first of many — see the top panel of Figure 3. On 1 June 1999, the 3-month eurodollar spot rate was at 5 per cent and the futures contract with delivery at September 1999 quoted at around $5\frac{3}{4}$; moreover, the *pdf* did show a negative skewness. Two weeks later the shift in the *pdf* mean and the decrease in the skewness show that markets were expecting an increase in short-term rates that was not signalled by the spot short rate quotes, which remained fairly steady. Two weeks after the increase of 0.25 percentage points market sentiment was slightly oriented towards additional increases (as shown by the shape of the *pdf* on 12 July and by its skewness) but the subsequent 1.5 point rise which took place by May 2000 was nor expected. The additional five increases in federal funds target rates neither shifted the *pdf*'s nor affected their skewness in the following month (not shown).

In the euro area the increase on 4 November 1999 was widely anticipated by agents — see the top panel of Figure 4.¹⁷ The *pdf* of 1 October 1999 had shown a large positive skewness slightly below 1.0 since the end of the summer. On 20 October the market perceptions were for a 3-month rate of around 3.75 per cent; one week after the increase perceptions were stable, if less dispersed around the mean. The skewness continued to remain large and pointed to additional rate increases.

In the United Kingdom the increase in the repo rate on 8 September 1999 was also anticipated by the markets: on 2 August 1999 the *pdf* was skewed to the left, assigning a larger probability to the occurrence of an increase in short-term rates — see the top panel of Figure 5. On 25 August the *pdf* shifted to the left, thus signalling an increase in the uncertainty of agents. One week after the rate increase the distribution moved markedly to the right, recording an additional increase in its skewness. Markets assigned a large likelihood to the occurrence of additional rate increases.

Phases of monetary easing

The starting date of cuts in US policy rates (FOMC on 3 January 2001) was partially anticipated by market participants. The *pdf*'s started to show a negative skewness (lower rates are deemed more probable) from the beginning of the last quarter of 2000 — see the bottom panel of Figure 3. Thus, the direction of short-term rates was expected even if the size of the decrease was not. On 15 December 2000, about three weeks before the Federal Reserve decision, the expected 3-month rate was below 6 per cent and the skewness was even more negative. After the decision had been taken, on 16 January 2001, market perception was of a substantial additional cut in policy rates (shown by a decrease in skewness as well as by the expected 3-month rate).

In the euro area, conversely, the monetary easing cycle on 10 May 2001 seems not to have been anticipated by the markets — see the bottom panel of Figure 4. In the course of the month preceding the GC meeting, from 4 to 25 April 2001, the *pdf* shifted to the right, even if the skewness decreased somewhat below zero. Moreover, one week after the decision had been taken, perceptions were almost unchanged.

 $^{^{17}}$ In this analysis I do not consider the possibility that the increase in the US policy rate could have affected the decisions in the euro area and the United Kingdom.

In the United Kingdom the decrease in the repo rate decided on 8 March 2001 was unexpected — see the bottom panel of Figure 5. The two pdfs of 1 and 23 February 2001 are identical; after the cut the dispersion increased substantially but the skewness remained steady around nil, signalling that the market was assigning the same probability to either move around the mean.

Results show that market participants did not expect the tightening monetary cycle in the United States and the additional policy-rate moves in 1999. The easing cycle of 2001 was largely anticipated and markets may have played a role in 'driving' US central bank decisions, since the distributions rapidly shifted to the left, showing that a business cycle slowdown was widely expected. In the euro area, agents had less clearer perceptions; the tightening cycle of 2001 was completely unexpected and, conversely, an increase seemed to be expected after the first rate cut. The uncertainty about the recovery in the euro area business cycle could have also contributed to this misperception. In the United Kingdom the rate increases of 1999 were not expected, the cuts only modified the uncertainty of market participants about successive monetary moves.

4.2 *A time series analysis*

In order to go beyond the simple comparison of pdfs at two dates around the central bank meeting, the time series of the pdf statistics have also been evaluated. These time series allow the dynamics of market sentiment during the sample period to be evaluated. I calculate the time series for the implied ATM volatility, the skewness, the kurtosis, the probability of a decrease of over 0.25 percentage points in the interest rate and that of an increase of over 0.25 percentage points, for the three areas.¹⁸ The skewness and the probability of an upward/downward move give the most straightforward intuition of the link between rate

¹⁸ The skewness is defined as the ratio between the third central moment and the standard deviation to the power of three, i.e. μ_3/σ^3 , the kurtosis as μ_4/σ^4 . The probability of a decrease in the interest rate by over 0.5 percentage points is defined as $\Pr(x - \overline{x} < -0.5) = \Phi(\overline{x} - 0.5)$, where x is the interest rate and \overline{x} is its average. Analogously, the probability of an increase of over 0.5 percentage points is defined as $\Pr(x - \overline{x} > 0.5) = 1 - \Phi(\overline{x} + 0.5)$.

changes and market expectations; the kurtosis consistently shows a strong co-movement with the skewness.

For the sake of brevity, I report the policy rate and the skewness in the three areas in Figure 6; the other statistics also document interesting features of pdfs. In the United States — top panel — the skewness gently increased by the second quarter of 1999, signalling that markets were expecting a monetary tightening; during the third quarter of 2000 the skewness rapidly moved downwards, anticipating the start of the monetary easing cycle. In the euro area — mid panel — the change in the skewness during 1999 were much more dramatic: it increased from -0.2 to 1.0 in two months, signalling that markets were unanimously expecting a policy-rate increase. The euro area easing cycle of 2001 seems not to have been anticipated by markets, which continued to show a zero skewness until January 2002. In the United Kingdom — bottom panel — the rate increase of 1999 was not anticipated, as shown by the skewness which lagged behind the policy-rate moves; conversely, the rapid decrease in skewness during the second half of 2000 shows that the tightening was largely expected.

The information content of the *pdf* statistics is evaluated by GMM-regressing the monthly change in the 3-month interest rate on a set of variables, namely

$$3M_t - 3M_{t-1} = \alpha_0 + \alpha_1 \cdot slope_{t-k} + \alpha_2 \cdot (3M - fut)_{t-k} + \alpha_3 \cdot (3M - 1M)_{t-k}$$

$$(4) \qquad \qquad +\alpha_4 \cdot skew_{t-k} + \alpha_5 \cdot vol_{t-k} + \varepsilon_t$$

for k = 1, 2, 3, 4, where 3M the 3-month rate, *slope* is difference between the 10- year government bond yield and the 3-month rate, 1M is the 1-month interest rate, *fut* is the futures contract written on the 3-month rate with the closest expiration date, *skew* the skewness of the *pdf*, *vol* the ATM implied volatility.

Results (coefficients and t-statistics are presented in Table 1) document that the slope of the yield curve is never significant, except in the euro area; in the euro area and in the UK the difference between the spot and the futures (3M - fut) is the most significant variable in explaining changes in short-term rates; the slope of the term structure at the very short-term (3M - 1M) is significant in all of the three markets to different degrees. Among the statistics obtained from the *pdf*s, the skewness is very significant in the euro area and in the UK, the volatility only in the euro area; neither in the US. The test on the joint significance of the skewness and of the volatility is never rejected at any lag in the euro area and in the UK.

	constant	slope	3M - fut	3M - 1M	skew	vol	R^2	Wald
				USA				
t-1	-0.09	0.02	-1.68	0.14	-0.02	0.03	0.59	0.70
	(-0.37)	(0.28)	(-10.12)	(2.15)	(-0.38)	(0.27)		
t-2	-0.33	0.06	-0.44	0.30	0.05	-0.09	0.28	0.34
	(-1.98)	(1.49)	(-2.68)	(3.55)	(0.95)	(-1.04)		
t-3	-0.45	0.12	0.62	1.48	-0.04	-0.18	0.21	0.40
	(-1.32)	(1.51)	(1.18)	(2.91)	(-0.83)	(-1.00)		
t-4	0.25	-0.05	-0.27	0.63	0.10	0.25	0.13	0.10
	(0.96)	(-0.96)	(-1.14)	(4.83)	(1.62)	(1.73)		
				euro area				
t-1	-0.17	0.08	-1.44	-0.09	-0.12	-0.05	0.63	0.05
	(-1.30)	(3.27)	(-7.10)	(-1.33)	(-2.42)	(-0.81)		
t-2	-0.39	0.07	-0.31	-0.36	0.21	-0.16	0.46	0.00
	(-3.01)	(2.33)	(-1.70)	(-5.20)	(4.91)	(-2.95)		
t-3	-0.86	0.16	0.78	-0.20	0.19	-0.38	0.36	0.00
	(-5.64)	(4.86)	(4.32)	(-2.45)	(3.02)	(-5.24)		
t-4	-0.63	0.12	0.89	0.27	0.10	-0.27	0.33	0.00
	(-5.66)	(4.13)	(5.56)	(4.24)	(1.60)	(-5.14)		
				UK				
t-1	0.01	0.02	-0.09	0.25	0.16	0.04	0.29	0.00
	(0.05)	(0.37)	(-0.37)	(1.57)	(4.25)	(0.36)		
t-2	-0.32	0.10	0.58	0.40	0.24	-0.16	0.34	0.02
	(-1.77)	(1.82)	(4.36)	(4.98)	(2.49)	(-1.41)		
t-3	0.30	-0.07	-0.27	0.07	0.17	0.21	0.12	0.01
	(1.09)	(-0.80)	(-2.24)	(1.28)	(2.81)	(1.21)		
t-4	0.51	-0.13	-0.13	0.28	0.16	0.36	0.12	0.04
	(2.20)	(-1.75)	(-0.75)	(2.19)	(2.30)	(2.41)		

ESTIMATES OF EQUATION (4)

Monthly data; coefficients and t-statistics (in brackets) of GMM estimates for the equation $3M_t - 3M_{t-1} = \alpha_0 + \alpha_1 slope + \alpha_2 (3M - fut)_{t-k} + \alpha_3 (3M - 1M)_{t-k} + \alpha_4 skew_{t-k} + \alpha_5 vol_{t-k} + \varepsilon_t$ for k = 1, 2, 3, 4, where 3M is the 3-month rate, slope the difference between the 10-year government bond yield and the 3-month rate, 1M the 1-month rate, fut the futures contract on the 3-month rate with the closest expiration date, skew the skewness of the pdf, vol the ATM implied volatility. Instruments are a constant and the same regressors with t - k and t - k - 1. I use the Hansen method to take into account overlapping observations. The column under the header *Wald* reports the p-value for the Wald test that $\alpha_4 = \alpha_5 = 0$.

Despite the small cross-section sample I also investigate the time series extracted from the pdfs around the monetary policy meeting. Figure 7 reports the cross-sectional average and median skewness from 40 business days before to 40 business days after the meetings of the monetary authorities in the three areas, gathered according to the directive announced; in what follows, a tightening/steady/easing meeting is defined as one where an increase/nochange/decrease is made. In the United States, the skewness around a tightening move is steady around 0.5 and does not show a particular trend; the average around a meeting where no change is made is steady before the meeting and slightly decreasing afterwards; the skewness around an easing move is increasing until the meeting and accelerates thereafter. The only clear trend is shown by the skewness around an easing meeting, which supports the hypothesis of a mean-reverting process for market expectations. In the euro area, there is also no trend in the skewness. The main difference with the US is that the skewness around an easing meeting has negative values of around -0.1. In the United Kingdom the skewness around an easing meeting shows an increasing trend after the meeting as in the United States; no trend appears in the other two lines.

In the three areas the skewness around a tightening meeting is largely positive, signalling that markets, on average, correctly predict an increase in short-term rates; it is either negative or nil around an easing meeting but reverts soon after. In order to assess market expectation movements around a meeting, I regress the cross-sectional average and median on a constant and a trend; results are shown in Table 5. The coefficients of the trend document that around a meeting where a decrease is made the skewness tends to move upwards, while the trend is uncertain around a tightening meeting; for a neutral meeting the positive value of the coefficient could be due to the extremely low level of interest rates during the sample period, which made it reasonable to assume an upward move in the foreseeable future.

Table 5 also reports the p-value for a test of equality of the means before and after monetary meetings of mean and median; results show that there is a significant difference between means in almost all of the cases.

A caveat is in order. The averages are calculated on very small samples (for example the increasing meetings for the United States are only six, the decreasing meetings eleven, the steady meetings thirteen) and confidence intervals are quite large. Only the confidence interval of average skewness — computed as plus and minus two standard errors — around a tightening meeting does not contain the zero level in all the three areas.¹⁹

¹⁹ The asymptotic distribution for the skewness and the kurtosis are, respectively, $\sqrt{n} \cdot skewness \stackrel{d}{\sim} N(0,6)$

5. Conclusion

The use of *pdfs* implied in futures options is extensively used by central banks to assess market perceptions on future monetary moves. The analysis of the *pdf* makes it possible to go beyond the simple point estimates provided by the futures contract for the chosen delivery date since it provides confidence intervals characterized by their skewness and probability of extreme values. This paper aims to evaluate whether this application has a solid rationale. The methodology applied to the monetary policy decisions taken in the United States, the euro area and the United Kingdom from January 1999 to May 2002 shows that markets correctly forecasted the monetary easing of 2001 in the United States in the course of the second half of 2000, but not for the euro area and the United Kingdom. The evidence for the tightening cycle of 1999 is mixed: markets expected an increase in euro area policy rates at the beginning of 1999; expectations were less clear for the United States' increases. In the case of the United Kingdom the increase was not forecasted.

A cross-sectional analysis might have wider possibility of application but it is still at an early stage given the small sample. However, preliminary results show that the time series statistics extracted from the pdfs have different levels and slopes according to the expected monetary policy decision.

and $\sqrt{n} \cdot kurtosis \stackrel{d}{\sim} N(3, 24)$, where n is the sample population.

Tables and figures

TABLE 2

Date Directive deposit main refinancing marginal lending facility operations facility fixed variable June 6, 2003 1.00 2.00 3.00 Decrease March 7, 2003 Decrease 1.50 2.50 3.50 December 6, 2002 Decrease 1.75 2.75 3.75 November 9, 2001 2.25 3.25 4.25 Decrease _ September 18, 2001 Decrease 2.75 3.75 4.75 _ August 31, 2001 3.25 4.25 5.25 Decrease _ May 11, 2001 3.50 4.50 5.50 Decrease October 6, 2000 Increase 3.75 4.75 5.75 _ September 1, 2000 Increase 3.50 4.50 5.50 _ June 28 (announced on 8 June*), 2000 3.25 4.25 5.25 Maintain _ June 9, 2000 Increase 3.25 4.25 5.25 _ April 28, 2000 Increase 2.75 3.75 4.75 _ March 17, 2000 2.50 3.50 4.50 Increase _ February 4, 2000 2.25 3.25 4.25 Increase _ November 5, 1999 2.00 3.00 4.00 Increase _ April 9, 1999 Decrease 1.50 2.50 3.50 _ January 22, 1999 Decrease 2.00 3.00 4.50 _ January 4, 1999 Maintain 2.75 3.00 3.25 _ January 1, 1999 Maintain 2.00 3.00 4.50 _

ECB DIRECTIVES AND POLICY INTEREST RATES

* Announcement of change from fixed to variable rate from the 28 June.

TABLE 3

Date	Directive	Funds rate	Discount rate	Official bias
June 25, 2003	Decrease	1.00	0.50	Easier
November 6, 2002	Decrease	1.25	0.75	Easier
March 19, 2002	Maintain	1.75	1.25	Neutral
January 29-30, 2002	Maintain	1.75	1.25	Easier
December 11, 2001	Decrease	1.75	1.25	Easier
November 6, 2001	Decrease	2.00	1.50	Easier
October 2, 2001	Decrease	2.50	2.00	Easier
September 17*, 2001	Decrease	3.00	2.50	Easier
August 21, 2001	Decrease	3.50	3.00	Easier
June 27, 2001	Decrease	3.75	3.25	Easier
May 15, 2001	Decrease	4.00	3.50	Easier
April 18*, 2001	Decrease	4.50	4.00	Easier
March 20, 2001	Decrease	5.00	4.50	Easier
January 31, 2001	Decrease	5.50	5.00	Easier
January 3-4*, 2001	Decrease	6.00	5.50	Easier
December 19, 2000	Maintain	6.50	6.00	Easier
June 27, 2000	Maintain	6.50	6.00	Inflationary
May 16, 2000	Increase	6.50	6.00	Inflationary
March 21, 2000	Increase	6.00	5.50	Inflationary
February 2, 2000	Increase	5.75	5.25	Tighter
December 21, 1999	Maintain	5.50	5.00	Neutral
November 16, 1999	Increase	5.50	5.00	Neutral
October 5, 1999	Maintain	5.25	4.75	Tighter
August 24, 1999	Increase	5.25	4.75	Neutral
June 29-30, 1999	Increase	5.00	4.50	Neutral
May 18, 1999	Maintain	4.75	4.50	Tighter
February 2-3, 1999	Maintain	4.75	4.50	Tighter

FOMC DIRECTIVES AND POLICY INTEREST RATES

* Denotes policy change outside of scheduled meeting

BANK OF ENGLAND DIRECTIVES AND POLICY INTEREST RATES

Date	Directive	repo rate
May 5-6, 2004	Increase	4.00
November 5-6, 2003	Increase	3.75
July 9-10, 2003	Decrease	3.50
February 5-6, 2003	Decrease	3.75
November 7-8, 2001	Decrease	4.00
October 3-4, 2001	Decrease	4.50
September 18, 2001	Decrease	4.75
August 1-2, 2001	Decrease	5.00
May 9-10, 2001	Decrease	5.25
April 4-5, 2001	Decrease	5.50
March 7-8, 2001	Decrease	5.75
February 9-10, 2000	Increase	6.00
January 12-13, 2000	Increase	5.75
November 3-4, 1999	Increase	5.50
September 7-8, 1999	Increase	5.25
June 9-10, 1999	Decrease	5.00
April 7-8, 1999	Decrease	5.25
February 3-4, 1999	Decrease	5.50
January 6-7, 1999	Decrease	6.00

Figure 3 CONSTANT 90-DAY HORIZON PDFs FOR US SHORT-TERM RATES

Around the start of US monetary tightening (FOMC of 29/30 June 1999)



Around the start of US monetary easing (FOMC of 3 January 2001)



To the extent that agents are risk averse their true pdf may

differ from those shown and may be shifted to the right.

The *pdf* indicates the likelihood of a particular event occurring.

The probability of the 3-month eurodollar deposit rate being

 $X\pm0.125$ ticks is given by the area under the curve between

X+0.125 and X-0.125. The area under the whole curve is always 100%.

Figure 4

CONSTANT 90-DAY HORIZON PDFs FOR EURO AREA SHORT-TERM RATES

Around the start of euro area monetary tightening (GC of 4 November 1999)



Around the start of euro area monetary easing (GC of 10 May 2001)



To the extent that agents are risk averse their true pdf may

differ from those shown and may be shifted to the right.

The *pdf* indicates the likelihood of a particular event occurring.

The probability of the 3-month eurodollar deposit rate being

 $\rm X\pm 0.125$ ticks is given by the area under the curve between

X+0.125 and X-0.125. The area under the whole curve is always 100%.

Figure 5 CONSTANT 90-DAY HORIZON PDFs FOR UK SHORT-TERM RATES

Around the start of UK monetary tightening (MPC of 7/8 September 1999)



Around the start of UK monetary easing (MPC of 7/8 March 2001)



To the extent that agents are risk averse their true pdf may

differ from those shown and may be shifted to the right.

The *pdf* indicates the likelihood of a particular event occurring.

The probability of the 3-month eurodollar deposit rate being

 $X\pm0.125$ ticks is given by the area under the curve between

X+0.125 and X-0.125. The area under the whole curve is always 100%.



Figure 6

SKEWNESS AROUND POLICY MEETINGS



The three lines show the mean and the median of the skewness 40 business days before and 40 after monetary policy meetings held at t = 0 according to the chosen directive; the dashed line presents the mean/median skewness around a meeting when an increase has been announced, the dotted line around a meeting when a decrease has been announced, the continuous line around a meeting when no change in the policy rate has been made.

TABLE 5

OF THE MEANS BEFORE AND AFTER THE MONETARY MEETING median mean tightening easing steady tightening easing steady USA -0.17** 0.38** 0.48** 0.07** 0.62** 0.36** constant 0.00** 0.00** -0.00 0.00** 0.00 0.00 trend 0.09 0.00 0.35 0.27 test p-value 0.00 0.03 0.53** 0.51** 0.15** 0.57** -0.10** -0.04** constant euro area 0.00** 0.00^{**} 0.00** -0.00** trend -0.00** 0.00 0.00 0.34 0.00 0.00 0.67 0.00 test p-value 0.22** UK 0.35** 0.41** 0.047** 0.40** constant 0.01 0.00** 0.00** -0.00 0.00** -0.00** 0.00 trend test p-value 0.96 0.00 0.01 0.35 0.00 0.01

CROSS-SECTIONAL REGRESSION AND TEST OF EQUALITY

The table reports coefficients and t-statistics of the regression of the cross sectional mean and median of Figure 7 on a constant and a trend. * (**) indicate that the coefficient is significant at the 5 (1) per cent level. The rows with the 'test p-value' report a test of equality of the mean: it is made by classifying the sample before and after the monetary meeting and is based on a single-factor, between-subjects analysis of variance (ANOVA); the row reports the p-value of a t-test with 79 degrees of freedom: the null hypothesis that the sample means are equal is rejected at α per cent for a p-value smaller than α .

Appendix

The data

Futures options are traded on the same exchanges as the underlying futures contract and are executed and cleared with similar procedures as the underlying, the main differences being the margining and resettlement practices. A distinction can be drawn between *pure* and *conventional* futures options (Duffie 1989). The former give the buyer daily any change in the futures option price in order to mark the buyer's margin account. The latter require the payment of the option premium when it is purchased and at exercise pays the buyer the difference between the underlying and the strike price. Thus, a *pure* futures option is not an option at all, but rather a futures contract that delivers the corresponding *conventional* option on expiration. The futures options traded at the London International Financial Futures and Options Exchange (LIFFE) have rulings that make them more like *pure* than *conventional* options. Futures options traded at CME are hybrids of these two categories. Thus, the *pure* futures option price is the forward price of the underlying option. If O(t, T, ...) is the European type *conventional* futures option price at time t with expiration at T, its *pure* futures option equivalent price is $O^F(t, T, ...) = e^{-r(T-t)} \cdot O(t, T, ...)$, where r is the risk-free rate.

I collect daily quotes of end-of-day settlement prices of futures options written on 3month euro-currency interbank deposits denominated in US dollars, euros and sterling from January 1999 to May 2004. US dollar 3-month interest-rate futures options are traded at the Chicago Mercantile Exchange (CME), while euro and sterling 3-month interest-rate futures options are traded at the LIFFE. Both datasets are available on the Internet at the addresses *www.cme.com* and *www.liffe.com*, as well as from Thomson Financial Datastream.

At LIFFE, the American-type options are written on 3-month interest-rate futures contracts on euro and sterling interbank deposits and have delivery months in March, June, September, December — quarterly expiry months — and two serial months, so that ten expiry months for the euro and six for the sterling are available for trading, with the nearest three expiry months being consecutive calendar months. The last trading day is two business days prior to the third Wednesday of the expiry month for both serial expiry months and quarterly expiry months. The minimum price movement is 0.005 and the associated tick sizes and values are euro 12.50 and £6.25 for the euro and sterling deposits, respectively. Strike price

intervals are 0.25 but 0.125 for the first four quarterly expiry months and for the serial expiry months. As regards futures contracts, they are written onto the European Bankers Federations' Euribor Offered Rate (EBF Euribor) for 3-month euro deposits and onto the British Bankers' Association London Interbank Offered Rate (BBA LIBOR) for 3-month sterling deposits; the futures contract delivery month associated with each option expiry month is March in respect of the January, February and March expiry months; June in respect of the April, May and June expiry months; September in respect of the July, August and September expiry months; December in respect of the October, November and December expiry months.

At CME, the American-type options are written on the IMM index for the 3-month 'Eurodollar Time Deposit' futures contracts and have delivery months in March, June, September, December — March quarterly cycle — and two serial months, so that eight expiry months area available for trading, with the nearest three expiry months being consecutive calendar months. The last trading day is two business days prior to the third Wednesday of the expiry month at 7.00 on the day of exercise for both quarterly cycle and serial expiry months. The strike price interval is 0.25 and 0.125 for some particular expiry months. There are three minimum price movements, depending on the expiry month, 0.0025, 0.005 and 0.01 (one tick) whose associated tick sizes and values are \$6.25, \$12.50 and \$25.00, respectively. The IMM index for 3-month 'Eurodollar Time Deposit' futures contracts underlying is the 3-month BBA LIBOR on eurodollar deposits, with a principal value of \$1,000,000, forty delivery months in the March quarterly cycle, and the four nearest serial contract months.

In what follows we refer to policy rates in the three areas to the target federal funds rate in the US, the rate on main refinancing operations in the euro area, the repurchase agreement rate in the UK. The 3-month interest rates are the 3-month EBF Euribor for euro-denominated interbank deposits, the three 3 BBA LIBOR for dollar and sterling-denominated deposits. policy-rate target changes are determined eight times per year by the Federal Open Market Committee (FOMC) in the United States, and every month by the General Council Meeting in the euro area,²⁰ every month by the Monetary Policy Committee (MPC) in the UK. Central bank directives and policy rates are shown in Tables 1-3.

 $^{^{20}\,}$ Since 8 November the General Council meeting where policy changes are decided has a four week frequency.

References

- Aït-Sahalia Y. and Lo A.W. (1998), "Non Parametric Estimation of State-Price Densities Implicit in Financial Asset Prices", *Journal of Finance*, 53(2), April, 499-547.
- Aït-Sahalia Y., Wang Y. and Yared F. (2001), "Do Option Markets Correctly Price the Probabilities of Movement of the Underlying Asset?" *Journal of Econometrics*, 102, 67-110.
- Amin K.I. and Ng V. K. (1997), "Inferring Future Volatility from the Information in Implied Volatility in Eurodollar Options: A New Approach", *The Review of Financial Studies*, 10(2), 332-67.
- Bank of International Settlements (1999), *Estimating and Interpreting Probability Density Functions*, proceedings of the workshop held at the BIS on 14 June 1999.
- Bahra B. (1996), "Probability Distributions of Future Asset Prices Implied by Option Prices," *Bank of England Quarterly Bullettin*, 36(3), 299-311.
- Bank of England (2004), Quarterly Bulletin, Spring.
- Barone-Adesi G. Whaley R.E., (1987), "Efficient Analytic Approximation of American Option Values," *Journal of Finance*, XLII(2), 301-320.
- Bliss R. and Panigirtzouglou M. (2002), "Testing the Stability of Implied Probability Density Functions," *Journal of Banking and Finance*, 26, 381-422.
- Bliss R. and Panigirtzouglou M. (2004), "Option-Implied Risk Aversion Estimates," *Journal* of Finance, LIX(1), February, 407-446.
- Campa J.M. And Chang K.P.H. (1997), "The Forecasting Ability of Correlations Implied in Foreign Exchange Options", NBER Working Paper, 5974, March.
- Clews R., Panigirtzouglou and Proudman J. (2000), "Recent Developments in Extracting Information From Option Markets", *Bank of England Quarterly Bullettin*, February, 50-60.
- Cooper N. (2000), "Testing Techniques for Estimating Implied Risk-Neutral Densities from the Prices of European and American-Style Options", *unpublished paper*, Bank of England.
- Coutant S. (1999), "Implied risk aversion in option prices using Hermite polynomials," in *Estimating and Interpreting Probability Density Functions*, proceedings of the workshop held at the BIS on 14 June 1999, November.

Deutsche Bundesbank (1996), Monthly Report, April.

Duffie D. (1989), Futures Markets, Prentice Hall, New Jersey.

- Fornari F., Violi R. (1998), "The Probability Density Function of Interest Rates Implied in the Price of Options", Banca d'Italia, Temi di Discussione, No. 339. October.
- Hull J.C. (2000), *Options, Futures, & Other Derivatives*, 4th edition, Prentice Hall International, New Jersey.
- Jondeau E., Rockinger M. (2000), "Reading the Smile: the Message Conveyed by Methods which Infer Risk Neutral Densities," *Journal of International Money and Finance*, 19, 885-915.
- Malz A.M. (1996), "Using Option Prices to Estimate Realignment Probabilities in the European Monetary System: the Case of Sterling-Mark", *Journal of International Money and Finance*, 15(5), 717-748.
- Malz A.M. (1997), "Estimating the Probability of Future Exchange Rates from Option Prices", *Journal of Derivatives*, Winter, 18-36.
- Melick W.M. and Thomas C.P. (1997), "Recovering an Asset's Implied PDF from Option Prices: An Application to Crude Oil during the Gulf Crisis", *Journal of Financial and Quantitative Analysis*, 32(1), March, 91-115.
- Neuhaus H. (1995), "The Information Content of Derivatives for Monetary Policy. Implied Volatilities and Probabilities", Discussion Paper 3/95, Economic Research Group of the Deutsche Bundesbank, July.
- Owens R.E. Webb R.H., (2001), "Using the Federal Funds Futures Market to Predict Monetary Policy Actions," *Business Economics*, 36, April, 44-48.
- Pericoli M. (2000), "Extracting Probability Density Functions from Options Prices," unpublished paper, Banca d'Italia, February.
- Söderlind P. Svensson L., (1997), "New Techniques to Extract Market Expectations From Financial Instruments", *Journal of Monetary Economics*, 40, 383-429.

- N. 520 *Errori di misura nell'indagine sui bilanci delle famiglie italiane*, by C. BIANCOTTI, G. D'ALESSIO and A. NERI (September 2004).
- N. 521 *Do mergers improve information? Evidence from the loan market*, by F. PANETTA, F. SCHIVARDI and M. SHUM (September 2004).
- N. 522 Tecnologia e dinamica dei vantaggi comparati: un confronto fra quattro regioni italiane, by C. BENTIVOGLI and F. QUINTILIANI (September 2004).
- N. 523 The short-term impact of government budgets on prices: evidence from macroeconometric models, by J. HENRY, P. HERNÀNDEZ DE Cos and S. MOMIGLIANO, (October 2004).
- N. 524 Pricing behavior and the comovement of productivity and labor: evidence from *firm-level data*, by D.J. MARCHETTI and F. NUCCI (December 2004).
- N. 525 Is there a cost channel of monetary policy transmission? An investigation into the pricing behaviour of 2,000 firms, by E. GAIOTTI and A. SECCHI (December 2004).
- N. 526 Foreign direct investment and agglomeration: Evidence from Italy, by R. BRONZINI (December 2004).
- N. 527 Endogenous growth in open economies: A survey, by A. F. Pozzolo (December 2004).
- N. 528 The role of guarantees in bank lending, by A. F. Pozzolo (December 2004).
- N. 529 *Does the ILO definition capture all unemployment*, by A. BRANDOLINI, P. CIPOLLONE and E. VIVIANO (December 2004).
- N. 530 *Household wealth distribution in Italy in the 1990s*, by A. BRANDOLINI, L. CANNARI, G. D'ALESSIO and I. FAIELLA (December 2004).
- N. 531 Cyclical asymmetry in fiscal policy, debt accumulation and the Treaty of Maastricht, by F. BALASSONE and M. FRANCESE (December 2004).
- N. 532 L'introduzione dell'euro e la divergenza tra inflazione rilevata e percepita, by P. DEL GIOVANE and R. SABBATINI (December 2004).
- N. 533 A micro simulation model of demographic development and households' economic behavior in Italy, by A. ANDO and S. NICOLETTI ALTIMARI (December 2004).
- N. 534 Aggregation bias in macro models: does it matter for the euro area?, by L. MONTEFORTE (December 2004).
- N. 535 Entry decisions and adverse selection: an empirical analysis of local credit markets, by G. GOBBI and F. LOTTI (December 2004).
- N. 536 An empirical investigation of the relationship between inequality and growth, by P. PAGANO (December 2004).
- N. 537 Monetary policy impulses, local output and the transmission mechanism, by M. CARUSO (December 2004).
- N. 538 An empirical micro matching model with an application to Italy and Spain, by F. PERACCHI and E.VIVIANO (December 2004).
- N. 539 La crescita dell'economia italiana negli anni novanta tra ritardo tecnologico e rallentamento della produttività, by A. BASSANETTI, M. IOMMI, C. JONA-LASINIO and F. ZOLLINO (December 2004).
- N. 540 Cyclical sensitivity of fiscal policies based on real-time data, by L. FORNI and S. MOMIGLIANO (December 2004).
- N. 541 L'introduzione dell'euro e le politiche di prezzo: analisi di un campione di dati individuali, by E. GAIOTTI and F. LIPPI (February 2005).
- N. 542 How do banks set interest rates?, by L. GAMBACORTA (February 2005).
- N. 543 Maxmin portfolio choice, by M. TABOGA (February 2005).
- N. 544 Forecasting output growth and inflation in the euro area: are financial spreads useful?, by A. NOBILI (February 2005).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi – Divisione Biblioteca e pubblicazioni – Via Nazionale, 91 – 00184 Rome (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- L. GUISO and G. PARIGI, *Investment and demand uncertainty*, Quarterly Journal of Economics, Vol. 114 (1), pp. 185-228, **TD No. 289** (November 1996).
- A. F. POZZOLO, *Gli effetti della liberalizzazione valutaria sulle transazioni finanziarie dell'Italia con l'estero*, Rivista di Politica Economica, Vol. 89 (3), pp. 45-76, **TD No. 296 (February 1997)**.
- A. CUKIERMAN and F. LIPPI, Central bank independence, centralization of wage bargaining, inflation and unemployment: theory and evidence, European Economic Review, Vol. 43 (7), pp. 1395-1434, TD No. 332 (April 1998).
- P. CASELLI and R. RINALDI, La politica fiscale nei paesi dell'Unione europea negli anni novanta, Studi e note di economia, (1), pp. 71-109, TD No. 334 (July 1998).
- A. BRANDOLINI, The distribution of personal income in post-war Italy: Source description, data quality, and the time pattern of income inequality, Giornale degli economisti e Annali di economia, Vol. 58 (2), pp. 183-239, TD No. 350 (April 1999).
- L. GUISO, A. K. KASHYAP, F. PANETTA and D. TERLIZZESE, Will a common European monetary policy have asymmetric effects?, Economic Perspectives, Federal Reserve Bank of Chicago, Vol. 23 (4), pp. 56-75, TD No. 384 (October 2000).

- P. ANGELINI, Are banks risk-averse? Timing of the operations in the interbank market, Journal of Money, Credit and Banking, Vol. 32 (1), pp. 54-73, **TD No. 266 (April 1996).**
- F. DRUDI and R: GIORDANO, *Default Risk and optimal debt management*, Journal of Banking and Finance, Vol. 24 (6), pp. 861-892, **TD No. 278 (September 1996)**.
- F. DRUDI and R. GIORDANO, *Wage indexation, employment and inflation,* Scandinavian Journal of Economics, Vol. 102 (4), pp. 645-668, **TD No. 292 (December 1996)**.
- F. DRUDI and A. PRATI, *Signaling fiscal regime sustainability*, European Economic Review, Vol. 44 (10), pp. 1897-1930, **TD No. 335 (September 1998)**.
- F. FORNARI and R. VIOLI, The probability density function of interest rates implied in the price of options, in: R. Violi, (ed.), Mercati dei derivati, controllo monetario e stabilità finanziaria, Il Mulino, Bologna, TD No. 339 (October 1998).
- D. J. MARCHETTI and G. PARIGI, Energy consumption, survey data and the prediction of industrial production in Italy, Journal of Forecasting, Vol. 19 (5), pp. 419-440, TD No. 342 (December 1998).
- A. BAFFIGI, M. PAGNINI and F. QUINTILIANI, Localismo bancario e distretti industriali: assetto dei mercati del credito e finanziamento degli investimenti, in: L.F. Signorini (ed.), Lo sviluppo locale: un'indagine della Banca d'Italia sui distretti industriali, Donzelli, TD No. 347 (March 1999).
- A. SCALIA and V. VACCA, Does market transparency matter? A case study, in: Market Liquidity: Research Findings and Selected Policy Implications, Basel, Bank for International Settlements, TD No. 359 (October 1999).
- F. SCHIVARDI, *Rigidità nel mercato del lavoro, disoccupazione e crescita*, Giornale degli economisti e Annali di economia, Vol. 59 (1), pp. 117-143, **TD No. 364 (December 1999)**.
- G. BODO, R. GOLINELLI and G. PARIGI, *Forecasting industrial production in the euro area*, Empirical Economics, Vol. 25 (4), pp. 541-561, **TD No. 370 (March 2000)**.
- F. ALTISSIMO, D. J. MARCHETTI and G. P. ONETO, *The Italian business cycle: Coincident and leading indicators and some stylized facts*, Giornale degli economisti e Annali di economia, Vol. 60 (2), pp. 147-220, **TD No. 377 (October 2000)**.

- C. MICHELACCI and P. ZAFFARONI, (*Fractional*) *Beta convergence*, Journal of Monetary Economics, Vol. 45, pp. 129-153, **TD No. 383 (October 2000)**.
- R. DE BONIS and A. FERRANDO, The Italian banking structure in the nineties: testing the multimarket contact hypothesis, Economic Notes, Vol. 29 (2), pp. 215-241, TD No. 387 (October 2000).

- M. CARUSO, Stock prices and money velocity: A multi-country analysis, Empirical Economics, Vol. 26 (4), pp. 651-72, TD No. 264 (February 1996).
- P. CIPOLLONE and D. J. MARCHETTI, *Bottlenecks and limits to growth: A multisectoral analysis of Italian industry*, Journal of Policy Modeling, Vol. 23 (6), pp. 601-620, **TD No. 314 (August 1997)**.
- P. CASELLI, Fiscal consolidations under fixed exchange rates, European Economic Review, Vol. 45 (3), pp. 425-450, TD No. 336 (October 1998).
- F. ALTISSIMO and G. L. VIOLANTE, Nonlinear VAR: Some theory and an application to US GNP and unemployment, Journal of Applied Econometrics, Vol. 16 (4), pp. 461-486, TD No. 338 (October 1998).
- F. NUCCI and A. F. POZZOLO, *Investment and the exchange rate*, European Economic Review, Vol. 45 (2), pp. 259-283, **TD No. 344 (December 1998)**.
- L. GAMBACORTA, On the institutional design of the European monetary union: Conservatism, stability pact and economic shocks, Economic Notes, Vol. 30 (1), pp. 109-143, **TD No. 356 (June 1999)**.
- P. FINALDI RUSSO and P. ROSSI, Credit costraints in italian industrial districts, Applied Economics, Vol. 33 (11), pp. 1469-1477, TD No. 360 (December 1999).
- A. CUKIERMAN and F. LIPPI, *Labor markets and monetary union: A strategic analysis*, Economic Journal, Vol. 111 (473), pp. 541-565, **TD No. 365 (February 2000)**.
- G. PARIGI and S. SIVIERO, An investment-function-based measure of capacity utilisation, potential output and utilised capacity in the Bank of Italy's quarterly model, Economic Modelling, Vol. 18 (4), pp. 525-550, TD No. 367 (February 2000).
- F. BALASSONE and D. MONACELLI, *Emu fiscal rules: Is there a gap?*, in: M. Bordignon and D. Da Empoli (eds.), Politica fiscale, flessibilità dei mercati e crescita, Milano, Franco Angeli, **TD No.** 375 (July 2000).
- A. B. ATKINSON and A. BRANDOLINI, Promise and pitfalls in the use of "secondary" data-sets: Income inequality in OECD countries, Journal of Economic Literature, Vol. 39 (3), pp. 771-799, TD No. 379 (October 2000).
- D. FOCARELLI and A. F. POZZOLO, The determinants of cross-border bank shareholdings: An analysis with bank-level data from OECD countries, Journal of Banking and Finance, Vol. 25 (12), pp. 2305-2337, TD No. 381 (October 2000).
- M. SBRACIA and A. ZAGHINI, *Expectations and information in second generation currency crises models*, Economic Modelling, Vol. 18 (2), pp. 203-222, **TD No. 391 (December 2000)**.
- F. FORNARI and A. MELE, Recovering the probability density function of asset prices using GARCH as diffusion approximations, Journal of Empirical Finance, Vol. 8 (1), pp. 83-110, TD No. 396 (February 2001).
- P. CIPOLLONE, La convergenza dei salari manifatturieri in Europa, Politica economica, Vol. 17 (1), pp. 97-125, **TD No. 398 (February 2001)**.
- E. BONACCORSI DI PATTI and G. GOBBI, The changing structure of local credit markets: Are small businesses special?, Journal of Banking and Finance, Vol. 25 (12), pp. 2209-2237, TD No. 404 (June 2001).
- G. MESSINA, Decentramento fiscale e perequazione regionale. Efficienza e redistribuzione nel nuovo sistema di finanziamento delle regioni a statuto ordinario, Studi economici, Vol. 56 (73), pp. 131-148, TD No. 416 (August 2001).

- R. CESARI and F. PANETTA, Style, fees and performance of Italian equity funds, Journal of Banking and Finance, Vol. 26 (1), TD No. 325 (January 1998).
- L. GAMBACORTA, Asymmetric bank lending channels and ECB monetary policy, Economic Modelling, Vol. 20 (1), pp. 25-46, **TD No. 340 (October 1998)**.
- C. GIANNINI, "Enemy of none but a common friend of all"? An international perspective on the lenderof-last-resort function, Essay in International Finance, Vol. 214, Princeton, N. J., Princeton University Press, TD No. 341 (December 1998).
- A. ZAGHINI, Fiscal adjustments and economic performing: A comparative study, Applied Economics, Vol. 33 (5), pp. 613-624, TD No. 355 (June 1999).
- F. ALTISSIMO, S. SIVIERO and D. TERLIZZESE, *How deep are the deep parameters?*, Annales d'Economie et de Statistique,.(67/68), pp. 207-226, **TD No. 354 (June 1999)**.
- F. FORNARI, C. MONTICELLI, M. PERICOLI and M. TIVEGNA, *The impact of news on the exchange rate of the lira and long-term interest rates*, Economic Modelling, Vol. 19 (4), pp. 611-639, **TD No. 358** (October 1999).
- D. FOCARELLI, F. PANETTA and C. SALLEO, *Why do banks merge?*, Journal of Money, Credit and Banking, Vol. 34 (4), pp. 1047-1066, **TD No. 361 (December 1999)**.
- D. J. MARCHETTI, Markup and the business cycle: Evidence from Italian manufacturing branches, Open Economies Review, Vol. 13 (1), pp. 87-103, **TD No. 362 (December 1999)**.
- F. BUSETTI, *Testing for stochastic trends in series with structural breaks*, Journal of Forecasting, Vol. 21 (2), pp. 81-105, TD No. 385 (October 2000).
- F. LIPPI, *Revisiting the Case for a Populist Central Banker*, European Economic Review, Vol. 46 (3), pp. 601-612, **TD No. 386 (October 2000)**.
- F. PANETTA, The stability of the relation between the stock market and macroeconomic forces, Economic Notes, Vol. 31 (3), TD No. 393 (February 2001).
- G. GRANDE and L. VENTURA, Labor income and risky assets under market incompleteness: Evidence from Italian data, Journal of Banking and Finance, Vol. 26 (2-3), pp. 597-620, TD No. 399 (March 2001).
- A. BRANDOLINI, P. CIPOLLONE and P. SESTITO, *Earnings dispersion, low pay and household poverty in Italy, 1977-1998*, in D. Cohen, T. Piketty and G. Saint-Paul (eds.), The Economics of Rising Inequalities, pp. 225-264, Oxford, Oxford University Press, **TD No. 427** (November 2001).
- L. CANNARI and G. D'ALESSIO, La distribuzione del reddito e della ricchezza nelle regioni italiane, Rivista Economica del Mezzogiorno (Trimestrale della SVIMEZ), Vol. XVI (4), pp. 809-847, Il Mulino, TD No. 482 (June 2003).

- F. SCHIVARDI, *Reallocation and learning over the business cycle*, European Economic Review, , Vol. 47 (1), pp. 95-111, **TD No. 345 (December 1998)**.
- P. CASELLI, P. PAGANO and F. SCHIVARDI, Uncertainty and slowdown of capital accumulation in Europe, Applied Economics, Vol. 35 (1), pp. 79-89, **TD No. 372 (March 2000).**
- P. ANGELINI and N. CETORELLI, *The effect of regulatory reform on competition in the banking industry*, Federal Reserve Bank of Chicago, Journal of Money, Credit and Banking, Vol. 35, pp. 663-684, **TD No. 380 (October 2000)**.

- P. PAGANO and G. FERRAGUTO, Endogenous growth with intertemporally dependent preferences, Contribution to Macroeconomics, Vol. 3 (1), pp. 1-38, **TD No. 382 (October 2000).**
- P. PAGANO and F. SCHIVARDI, *Firm size distribution and growth*, Scandinavian Journal of Economics, Vol. 105 (2), pp. 255-274, **TD No. 394 (February 2001)**.
- M. PERICOLI and M. SBRACIA, A Primer on Financial Contagion, Journal of Economic Surveys, Vol. 17 (4), pp. 571-608, TD No. 407 (June 2001).
- M. SBRACIA and A. ZAGHINI, *The role of the banking system in the international transmission of shocks*, World Economy, Vol. 26 (5), pp. 727-754, **TD No. 409 (June 2001)**.
- E. GAIOTTI and A. GENERALE, Does monetary policy have asymmetric effects? A look at the investment decisions of Italian firms, Giornale degli Economisti e Annali di Economia, Vol. 61 (1), pp. 29-59, TD No. 429 (December 2001).
- L. GAMBACORTA, The Italian banking system and monetary policy transmission: evidence from bank level data, in: I. Angeloni, A. Kashyap and B. Mojon (eds.), Monetary Policy Transmission in the Euro Area, Cambridge, Cambridge University Press, TD No. 430 (December 2001).
- M. EHRMANN, L. GAMBACORTA, J. MARTÍNEZ PAGÉS, P. SEVESTRE and A. WORMS, *Financial systems and the role of banks in monetary policy transmission in the euro area*, in: I. Angeloni, A. Kashyap and B. Mojon (eds.), Monetary Policy Transmission in the Euro Area, Cambridge, Cambridge University Press, **TD No. 432 (December 2001)**.
- F. SPADAFORA, Financial crises, moral hazard and the speciality of the international market: further evidence from the pricing of syndicated bank loans to emerging markets, Emerging Markets Review, Vol. 4 (2), pp. 167-198, TD No. 438 (March 2002).
- D. FOCARELLI and F. PANETTA, Are mergers beneficial to consumers? Evidence from the market for bank deposits, American Economic Review, Vol. 93 (4), pp. 1152-1172, **TD No. 448 (July 2002)**.
- E.VIVIANO, Un'analisi critica delle definizioni di disoccupazione e partecipazione in Italia, Politica Economica, Vol. 19 (1), pp. 161-190, **TD No. 450** (July 2002).
- M. PAGNINI, Misura e Determinanti dell'Agglomerazione Spaziale nei Comparti Industriali in Italia, Rivista di Politica Economica, Vol. 3 (4), pp. 149-196, **TD No. 452 (October 2002)**.
- F. BUSETTI and A. M. ROBERT TAYLOR, Testing against stochastic trend and seasonality in the presence of unattended breaks and unit roots, Journal of Econometrics, Vol. 117 (1), pp. 21-53, TD No. 470 (February 2003).

- F. LIPPI, Strategic monetary policy with non-atomistic wage-setters, Review of Economic Studies, Vol. 70 (4), pp. 909-919, TD No. 374 (June 2000).
- P. CHIADES and L. GAMBACORTA, *The Bernanke and Blinder model in an open economy: The Italian case*, German Economic Review, Vol. 5 (1), pp. 1-34, **TD No. 388 (December 2000)**.
- M. BUGAMELLI and P. PAGANO, *Barriers to Investment in ICT*, Applied Economics, Vol. 36 (20), pp. 2275-2286, **TD No. 420 (October 2001)**.
- A. BAFFIGI, R. GOLINELLI and G. PARIGI, *Bridge models to forecast the euro area GDP*, International Journal of Forecasting, Vol. 20 (3), pp. 447-460, **TD No. 456 (December 2002)**.
- D. AMEL, C. BARNES, F. PANETTA and C. SALLEO, Consolidation and Efficiency in the Financial Sector: A Review of the International Evidence, Journal of Banking and Finance, Vol. 28 (10), pp. 2493-2519, TD No. 464 (December 2002).
- M. PAIELLA, *Heterogeneity in financial market participation: appraising its implications for the C-CAPM*, Review of Finance, Vol. 8, pp. 1-36, **TD No. 473 (June 2003)**.
- E. BARUCCI, C. IMPENNA and R. RENÒ, *Monetary integration, markets and regulation*, Research in Banking and Finance, (4), pp. 319-360, **TD No. 475** (June 2003).

- E. BONACCORSI DI PATTI and G. DELL'ARICCIA, *Bank competition and firm creation*, Journal of Money Credit and Banking, Vol. 36 (2), pp. 225-251, **TD No. 481 (June 2003)**.
- R. GOLINELLI and G. PARIGI, Consumer sentiment and economic activity: a cross country comparison, Journal of Business Cycle Measurement and Analysis, Vol. 1 (2), pp. 147-172, TD No. 484 (September 2003).
- L. GAMBACORTA and P. E. MISTRULLI, *Does bank capital affect lending behavior?*, Journal of Financial Intermediation, Vol. 13 (4), pp. 436-457, **TD No. 486 (September 2003)**.
- F. SPADAFORA, Il pilastro privato del sistema previdenziale: il caso del Regno Unito, Rivista Economia Pubblica, (5), pp. 75-114, TD No. 503 (June 2004).
- G. GOBBI e F. LOTTI, Entry decisions and adverse selection: an empirical analysis of local credit markets, Journal of Financial services Research, Vol. 26 (3), pp. 225-244, TD No. 535 (December 2004).
- F. CINGANO e F. SCHIVARDI, *Identifying the sources of local productivity growth*, Journal of the European Economic Association, Vol. 2 (4), pp. 720-742, **TD No. 474 (June 2003)**.
- C. BENTIVOGLI e F. QUINTILIANI, Tecnologia e dinamica dei vantaggi comparati: un confronto fra quattro regioni italiane, in C. Conigliani (a cura di), Tra sviluppo e stagnazione: l'economia dell'Emilia-Romagna, Bologna, Il Mulino, **TD No. 522** (October 2004).

- A. DI CESARE, Estimating Expectations of Shocks Using Option Prices, The ICFAI Journal of Derivatives Markets, Vol. II (1), pp. 42-53, TD No. 506 (July 2004).
- M. OMICCIOLI, Il credito commerciale: problemi e teorie, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 494 (June 2004).
- L. CANNARI, S. CHIRI e M. OMICCIOLI, Condizioni del credito commerciale e differenzizione della clientela, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 495 (June 2004).
- P. FINALDI RUSSO e L. LEVA, Il debito commerciale in Italia: quanto contano le motivazioni finanziarie?, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 496 (June 2004).
- A. CARMIGNANI, Funzionamento della giustizia civile e struttura finanziaria delle imprese: il ruolo del credito commerciale, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 497 (June 2004).
- G. DE BLASIO, Does trade credit substitute for bank credit?, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 498 (June 2004).
- M. BENVENUTI e M. GALLO, Perché le imprese ricorrono al factoring? Il caso dell'Italia, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 518 (October 2004).